Author's response to reviews

Title: Development of a comprehensive list of criteria for evaluating consumer education materials on colorectal cancer screening

Authors:

Maren Dreier (dreier.maren@mh-hannover.de)
Birgit Borutta (Borutta.birgit@mh-hannover.de)
Seidel Gabriele (Seidel.gabriele@mh-hannover.de)
Kreusel Inga (Kreusel.inga@mh-hannover.de)
Juergen Toeppich (juergen.toeppich@bzga.de)
Eva M Bitzer (bitzer@iseg.org)
Marie-Luise Dierks (Dierks.marie-luise@mh-hannover.de)
Ulla Walter (Walter.ulla@mh-hannover.de)

Version: 2 Date: 16 April 2013

Author's response to reviews:

Response to concern 1. Please document within the methods section of your manuscript the specific name of the organization that granted ethical approval to your study going ahead.

> We added the following sentence: “The study protocol was approved by the ethics committee of Hannover Medical School (Application No. 1803-2013).”